Operational beams \/

—>SFTPRO:
o Working now with 100/50/60 sharing
o High transmission and low losses at extraction

o PS colleagues worked on improving emittance share between islands and core (triggered by
observations from NA62) = better now? Unfortunately, we cannot measure this in the SPS

o Main issue so far on H4 R22-11 and R22-12 bends — it seems solved now

o Today at 15h00 test of EBC in operation (about 2h for setting up and then running normally) — there
was a significant work on RF and ABT to make this possible

—>HiRadMat:

o Very good progress on the experiment: almost fully completed — today access and tomorrow last
few shots

—>LHC:

o Attempt to increase intensity today

o Agreed to have half a day to take the high intensity LHC cycle for 4-5h during the day today as
HiRadMat won’t take beam



